Author’s response to reviews

Title: Inclusion of joint laxity, recurrent patellar dislocation, and short distal ulnae as a feature of Van Den Ende-Gupta syndrome: A case report

Authors:

Mohammad Al-Qattan (moqattan@hotmail.com)
Doaa Andejani (dandejani@hotmail.com)
Nadia Sakati (nasakati@gmail.com)
Khushnooda Ramzan (kramzan@kfshrc.edu.sa)
Faiqa Imtiaz (fahmad@kfshrc.edu.sa)

Version: 2 Date: 13 Jan 2018

Author’s response to reviews:

MGTC-D-17-00267R1

Inclusion of joint laxity, recurrent patellar dislocation, and short distal ulnae as a feature of Van Den Ende-Gupta syndrome: A case report

Dear Editor,

Thank you for provisionally accepting the paper. We have revised it as follows:

1. We have noted that you have two Keywords sections, please remove one so that there is just one Keywords section.

Response: Done

2. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

Response: Done. See lines 204-207
3. We note that you have not included an acknowledgements section. If you have no acknowledgements please put ‘Not Applicable’ in this section.

Response: Done. See lines 208-9

4. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Response: Done. See lines 183-4

5. Please remove the supplementary file which you have called "data availability". Please include all required information as to how data can be acquired in the Availability of data and materials section only.

Response: Done.

6. Please provide better quality images of the patients' X-rays. At the moment, the start menu can be seen in one image, and they appear to be screenshots of the X-rays. Please try to provide the X-ray image only.

Response: Done. See revised figures

7. Please remove any identifying writing from the X-ray images, such as dates and names, in order to protect the patients' anonymity.

Response: Done. See revised figures

8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Done.